Author’s response to reviews

Title: Childbirth assistance in Catalonia. An analysis of obstetric interventions in two groups of hospitals

Authors:

Ramon Escuriet (rescuriet@catsalut.cat)
Maria Pueyo (mjpuveyo@catsalut.cat)
Herminia Biescas (hbiescas@catsalut.cat)
Cristina Colls (cristinacolls@gencat.cat)
Isabel Espiga (iespiga@msssi.es)
Joanna White (jowhite67@yahoo.co.uk)
Josep Fuste (jfuste@catsalut.cat)
Vicente Ortun (vicente.ortun@upf.edu)

Version: 2 Date: 30 July 2013

Author’s response to reviews:

Description of the changes made:

1. Included an Ethics statement in the methodology section
2. Reference citations reviewed